Author's Response To Reviewer Comments

We thank the reviewers for their thoughtful comments, which helped us improve our study and the manuscript.

We have attached a detailed response to the reviews (Review_response_untracked.pdf) as well as a version of the updated manuscript with all changes from the original submission highlighted (diff_manuscript.pdf).